Author’s response to reviews

Title: Emotional stimuli candidates for behavioural intervention in the prevention of early childhood caries: a pilot study

Authors:

Michaela Bartosova (michaela.bartosova@fnusa.cz)
Miroslav Svetlak (miroslav.svetlak@fnbrno.cz)
Martina Kukletova (makukl@med.muni.cz)
Petra Borilova Linhartova (plinhart@med.muni.cz)
Lydie Izakovicova Holla (holla@med.muni.cz)

Version: 1 Date: 05 Jun 2018

Author’s response to reviews:

BBMC Public Health

Editor

Natalie Pafitis

Brno, June 5, 2018


authors: Michaela Bartosova, Miroslav Svetlak, Martina Kukletova, Petra Borilova Linhartova, Lydie Izakovicova Holla

Dear Editor,

We corrected our manuscript according to your suggestions from June 1, 2018.
• Our results are original and they are not presently under consideration for publication elsewhere
• They were conducted by the highest principles of human subjects
• Author contributions: MK, MS and MB designed the study, prepared the questionnaire and drafted the manuscript. MB examined the infants and addressed mothers for a participation in the study. MS, PBL and LIH performed statistical analysis and participated in the interpretation of data. PBL and LIH assisted with reviewing and revising the final manuscript. All authors approved the manuscript as submitted and take full responsibility for the manuscript.
• We enclose confirmation about review of article for BMC journal for potential discount from publication fee.
• We confirm that written informed consent was obtained from all participants. The imagines and photographs in Fig. 2 and Appendix 1 are ours and we have the consent for publication from the persons/guardians in the pictures.

Yours sincerely,

Lydie Izakovicova Holla, MD., Ph.D.
Professor
Clinic of Stomatology
and Department of Pathophysiology
Medical Faculty, Masaryk University
Pekarska 53, 656 91 Brno
Czech Republic
Fax: +420-543183406
e-mail: holla@med.muni.cz